Author's response to reviews

Title: The Efficacy and Safety Study of Electro-acupuncture for Severe Chronic Functional Constipation Study protocol for a Multicenter randomized controlled trial

Authors:

Zhishun Liu (liuzhishun@yahoo.com.cn)
Jia Liu (marie_liujia@yahoo.com.cn)
Ye Zhao (zakzy@163.com)
Yuying Cai (caiyuying@yahoo.com.cn)
Liyun He (hely3699@163.com)
Huanfang Xu (ziwantcm@126.com)
Xiaohua Zhou (azhou@u.washington.edu)
Shiyan Yan (yanshiyan0927@sina.com)
Lixing Lao (llao@compmmed.umm.edu)
Baoyan Liu (baoyan.liu.cacms@gmail.com)

Version: 2 Date: 21 April 2013

Author's response to reviews:

point-to-point answer to the comments